**Original Article** 



# 'Who, When, How to Share': Pilot study of a new disclosure decision-making programme for autistic adults

Autism I–I5 © The Author(s) 2024

Article reuse guidelines: sagepub.com/journals-permissions DOI: 10.1177/13623613231221685 journals.sagepub.com/home/aut **S Sage** 

# Emeline Han<sup>1</sup>, Katrina Scior<sup>2</sup>, Kana Grace<sup>1</sup>, Eric Heath<sup>3</sup>, Simone Dufresne<sup>4</sup> and Laura Crane<sup>1</sup>

### Abstract

In this pilot study, we report on the feasibility, acceptability and potential impact of 'Who, When, How to Share', a new disclosure decision-making programme for autistic adults. We evaluated this programme using pre-post mixed-methods surveys that included questions about participants' experiences of the programme and four psychosocial outcome measures. While 32 autistic adults completed the pre-programme survey and enrolled into the programme, there was high attrition, with only 19 autistic adults completing the programme and post-programme survey. We found that it was feasible to deliver the programme in a guided self-help format, although a minority of participants (who completed the programme independently and within the study time frame. High levels of satisfaction and positive qualitative feedback also suggested that the programme was acceptable to autistic adults. Participants provided useful recommendations for improvement, such as more interactive elements to enhance engagement. The programme showed potential to improve decisional conflict, disclosure-related distress, stigma-related stress and internalised stigma among autistic adults, though further evaluation of the impact of the programme is needed.

### Lay abstract

'Who, When, How to Share' is a new programme that aims to support autistic adults in making decisions around sharing their autistic identity with others. The programme involves working through a self-help guide independently over 3 weeks with optional peer support. We wanted to find out if autistic adults would join the programme and find it useful. Thirty-two autistic adults took part in the programme and 19 of them completed it. Most participants who completed the programme liked the programme and found it helpful, but some felt that they needed more time and support to complete it. They suggested that the programme would be more accessible if it was more interactive, such as including videos and other ways to gain feedback on their progress. Surveys filled in by participants before and after the programme suggested that they became more confident and less stressed about sharing their autistic identity with others, but some felt they still needed to build more confidence in order to handle negative attitudes from others. More work is needed to improve and test the programme further.

### Keywords

autism, autistic adults, decision-making, disclosure, feasibility, intervention, pilot, programme, stigma, support

<sup>1</sup>Centre for Research in Autism and Education (CRAE), IOE, UCL's Faculty of Education and Society, UK

<sup>2</sup>UCL Research Department of Clinical, Educational and Health Psychology, UK <sup>4</sup>Eliot-Pearson Department of Child Study & Human Development, Tufts University, USA

#### **Corresponding author:**

Emeline Han, Centre for Research in Autism and Education (CRAE), IOE, UCL's Faculty of Education and Society, 55-59 Gordon Square, London WCIH 0NU, UK. Email: emeline.han.16@ucl.ac.uk

<sup>&</sup>lt;sup>3</sup>Independent Consultant, UK

# Introduction

Autistic adults often face a dilemma of whether and how much to conceal or reveal about their autistic identity. Research from the perspective of non-autistic people suggests that disclosure generally leads to more positive attitudes and behavioural intentions towards autistic people (Flower et al., 2021; McMahon et al., 2021; Sasson & Morrison, 2019). However, research eliciting autistic perspectives reveals mixed outcomes - disclosure can sometimes lead to increased understanding and support, but other times lead to judgement and discrimination (Huang et al., 2022; Romualdez et al., 2021; Thompson-Hodgetts et al., 2020). Conversely, concealment and camouflaging may offer some protection from public stigma, but have also been associated with internalised stigma, stress and anxiety among autistic adults (Botha & Frost, 2020; Cage & Troxell-Whitman, 2019; Cook et al., 2021). Consequently, disclosing an autistic identity is a complex personal decision in which the risks and benefits have to be weighed carefully depending on context. A recent study found that autistic adults' disclosure decisions are influenced by a range of external and internal factors, including how safe they feel in a particular environment, how much energy they have at the time and what they hope to achieve by disclosing (Love et al., 2023).

To improve disclosure experiences, it is essential to reduce public stigma and create more inclusive environments, as well as empower autistic people to make informed and strategic disclosure decisions. Autism acceptance training programmes for non-autistic people have been developed and evaluated (Gillespie-Lynch et al., 2022; Jones et al., 2021), with ongoing efforts to improve both the programmes and their evidence base. While there are a number of support programmes for autistic people that include a segment on disclosure (Crane et al., 2021; Laugeson et al., 2011), none have specifically focused on disclosure. Outside of the autism field, a promising intervention to support disclosure decision-making is Honest, Open, Proud (HOP) (Corrigan et al., 2013). The original HOP programme was delivered in a peer-led group format for adults with mental health conditions, and it comprised three main sessions. Session 1 guided participants in challenging hurtful self-beliefs and evaluating the pros and cons of disclosure in different settings. Session 2 introduced participants to different ways of disclosing and taught them to judge how others may respond before disclosing. Session 3 supported participants in crafting empowering personal narratives as a basis for potential disclosure. HOP's theory of change is that supporting disclosure decisions can reduce stress associated with stigma and internalised stigma, thus contributing towards improved well-being (Scior et al., 2020). A recent meta-analysis of five randomised controlled trials showed that HOP had a positive impact on stigma stress and internalised stigma for people with

mental health conditions, while its impact on depression was less consistent (Rüsch & Kösters, 2021). HOP has been adapted for different populations, including people with Tourette's syndrome and people with dementia. Given that disclosure decisions could vary based on disability type, we recognised the importance of involving autistic people in determining whether and how HOP could be suitably adapted for autistic people.

To inform our potential adaptation of HOP, we conducted a consultation survey with 124 autistic adults as primary stakeholders and 38 parents/caregivers of autistic people as secondary stakeholders (Han et al., 2023). The survey presented information on the premise and goal of HOP, the original target group and delivery format, as well as a summary of the content of its three main sessions, to elicit feedback prior to any proposed adaptation. Both autistic adults and parents/caregivers thought it was important for autistic adults to have support in managing disclosure and stigma, and that it would be suitable to adapt HOP for autistic adults. However, concerns were raised that the programme may place an onus on individuals to disclose or challenge stigma. For the programme to be helpful, respondents stressed that it should be positive (i.e. promote self-acceptance) and practical (i.e. teach context-specific strategies for disclosure). Concerns were also raised about the group format of the original HOP programme, which could be challenging and inaccessible for some autistic people, especially those who were worried about disclosure or stigma. Given the heterogeneity of the autistic population, flexibility and choice were highlighted as key facilitators to taking part in such a programme. In particular, autistic adults preferred the programme to be led by autistic peers with shared experience, who would be best placed to understand and accommodate various needs (Han et al., 2023).

Based on the above feedback, we adapted HOP in a guided self-help format supplemented by optional peer support, that is, with the option to contact a trained autistic facilitator and/or join an online forum with other participants. We named this adapted programme 'Who, When, How to Share', as the goal of the intervention is not to persuade individuals to disclose, but rather to guide them in reaching safe and personally meaningful disclosure decisions. Throughout the original material (Corrigan et al., 2017; Wisconsin Initiative for Stigma Elimination & Corrigan, 2018), references to mental health challenges and the journey to recovery were replaced with appropriate language discussing an autism diagnosis/autistic identity and the journey to acceptance. Participants were encouraged to craft a personal narrative based on the neurodiversity paradigm (Kapp et al., 2013; Walker, 2021), including identifying and describing their strengths and interests as an autistic person, as well as their challenges and needs. General pros and cons of disclosing a mental health condition were replaced with autism-specific pros and cons of

Table I	•	Content	of	self-help	guide.
---------	---	---------	----	-----------	--------

Content
<ul><li>I.1. Weighing the pros and cons of disclosure in different settings</li><li>I.2. Weighing the pros and cons of different levels of disclosure</li></ul>
<ul><li>2.1. Drafting your personal story</li><li>2.2. Choosing what to share from your story</li><li>2.3. Replacing hurtful self-beliefs in your story</li></ul>
<ul><li>3.1. Identifying a good person to share with</li><li>3.2. Being prepared with responses and support</li><li>3.3. Putting it all together</li></ul>

disclosing in different contexts (e.g. to family/friends, educators/employers and public service providers). Quotes from autistic people were taken from existing research studies to ensure that examples were grounded in autistic people's lived experiences. A table detailing and explaining the key changes we made to each section of the guide is available in Supplementary File 1.

The Medical Research Council (MRC) outlines four phases in intervention research: (1) development, (2) feasibility testing, (3) evaluation and (4) implementation (Skivington et al., 2021). While traditionally overlooked, the MRC emphasises that feasibility testing is a valuable step to guide decisions on whether and how to proceed to the evaluation phase. Therefore, the primary aim of this study was to assess the feasibility and acceptability of the 'Who, When, How to Share' programme. A secondary aim was to explore the potential impact of this new programme on psychosocial outcomes for autistic adults.

# Methods

This study was conducted according to the MRC framework for developing and evaluating complex interventions (Skivington et al., 2021). A single-arm, pre-post design was employed. Participants completed a mixed-methods survey on Qualtrics within the 2 weeks prior to the start of the programme and within 2 weeks of completion of the programme (a copy of the survey is available in Supplementary File 2). Ethical approval was obtained via the Department of Psychology and Human Development at IOE, UCL's Faculty of Education and Society.

# Participants

Eligible participants were adults over the age of 18 in the United Kingdom who either had a formal diagnosis of autism or self-identified as autistic.<sup>1</sup> The study was first advertised to participants from our consultation survey (Han et al., 2023), followed by social media advertising. We aimed to recruit 20–30 autistic adults, following the rule of thumb for sample sizes in pilot studies (Julious, 2005; Lancaster et al., 2004).

### Intervention

Participants were allocated to two iterations of the programme that ran in September and November 2022, respectively.<sup>2</sup> All participants received a PDF of the full self-help guide and worksheets from the guide in Word format (available on the Open Science Framework at https://osf.io/vhry3/). We recommended that participants work through the self-help guide within 3 weeks corresponding to the three core sections of content: (1) when to share, (2) how to share and (3) who to share with (Table 1). We estimated that each section would take about 60-90 min to complete, but participants could work at their own pace. An example of a worksheet in the guide is shown in Figure 1. During the 3-week period, participants received weekly check-in emails from their assigned facilitators and were also given access to an optional online forum on Slack that was moderated by facilitators. The forum was organised into five 'channels': a general channel for introductions, programme-wide questions, announcements and clarifications, three separate channels to discuss thoughts and reflections specific to the three content sections, and a resources channel for sharing any additional resources outside of but related to the programme. Facilitators were the autistic members of our research team who had previous experience supporting autistic peers. Facilitators underwent a 3-h training session focused on checking understanding of fundamental concepts in the self-help guide, filling in key worksheets and discussing their role in the programme.

# Measures

*Primary outcome measures.* Feasibility of the intervention was assessed by (1) rates of recruitment and retention, including any given reasons for not participating or dropping out, and (2) levels of reported engagement (in the post-programme survey) and observed engagement (based on emails to facilitators and posts on the online forum), including any given reasons for not engaging. Acceptability was assessed via the post-programme survey, including (1) quantitative data on levels of satisfaction with and perceived helpfulness of the programme and



Figure 1. Example of worksheet in the guide.

(2) qualitative data on participants' expectations of the programme and whether these expectations were met, what they liked most/least about the programme, as well as suggestions for improvement.

Secondary outcome measures. Potential impact of the intervention was assessed quantitatively using four psychosocial outcome measures identified from previous evaluations of HOP, complemented by qualitative data on perceived impact of the programme. Autistic team members were involved in ensuring that the chosen measures were suitable and/or advising how they needed to be adapted for an autistic population. Adapted versions of the following measures were included in the pre- and post-intervention surveys, alongside free-text questions asking participants whether they found the measures appropriate and easy to complete (to inform any future trials of the programme) and whether the programme had changed their thoughts on disclosure and stigma.

Decisional Conflict Scale. The scale contains five subscales measuring perceptions of feeling: (1) uninformed, (2) unclear about personal values, (3) unsupported in decision-making, (4) uncertain about choosing options and (5) ineffective at decision-making (O'Connor, 1995). The 16 items are scored on a 5-point Likert-type scale (0 – strongly agree to 4 – strongly disagree). We retained all items in their original wording but added a preface to the scale clarifying that it pertained to a decision on whether or not to share an autism diagnosis/autistic identity with others. Overall scores and subscale scores were expressed as a percentage of the maximum possible score, ranging from 0 (no decisional conflict) to 100 (extremely high decisional conflict). The scale was previously validated (O'Connor, 1995), and internal consistency was found to be excellent<sup>3</sup> in this study (Cronbach's  $\alpha$ =0.963 pre-test,  $\alpha$ =0.940 post-test).

Disclosure-Related Distress Scale. The original scale asked,

In general, how comfortable would you feel talking to a [item 1: friend or family member; item 2: teacher or employer] about your mental health, for example, telling them you have a mental health diagnosis and how it affects you? (Mulfinger et al., 2018)

We replaced references to 'mental health' with 'autism' and 'teacher' with 'educator', given that our target group was autistic adults. We also added a third item on disclosing to a public service provider (e.g. health professional, police officer) to correspond with the three disclosure contexts addressed in our programme. Each item was scored on a 7-point Likert-type scale (1 – not at all to 7 – very much). It was not possible to appraise the psychometric quality of the scale as it is an unvalidated measure previously used as a screening tool for HOP (Mulfinger et al., 2018), and it was expected that participants would answer each item differently.

Stigma Stress Scale. This 8-item scale contains two subscales, with four items measuring perceived harm caused by stigma and four items measuring perceived resources to cope with such harm (Rüsch et al., 2009). We replaced references to 'people with mental illness' with 'autistic people', and 'prejudice' with 'stigma' to maintain consistency in terminology. Each item was scored on a 5-point Likert-type Scale (1 – strongly disagree to 5 – strongly agree), and an overall stress appraisal score was calculated by subtracting perceived coping resources from perceived harm. Internal consistency of the scale was good in this study (Cronbach's  $\alpha$ =0.836 pre-test,  $\alpha$ =0.851 post-test).

Internalised Stigma of Mental Illness. Internalised Stigma of Mental Illness (IMSI-10) is the brief, 10-item version of the Internalised Stigma of Mental Illness scale, measuring alienation, discrimination experience, social withdrawal, stereotype endorsement and stigma resistance (Boyd et al., 2014). We replaced references to 'mental illness' with 'autism' and 'mentally ill people' with 'autistic people'. Item 3, 'I don't socialize as much as I used to' was changed to 'I don't socialize as much as I would like to', as autism is a lifelong condition. Item 9, 'I can have a good, fulfilling life, despite my mental illness' was changed to 'I can have a good, fulfilling life as an autistic person', as there may be people who feel they can have a good life because of their autism or cannot distinguish what is because of or despite, since autism is an intrinsic part of them. Each item was scored on a 5-point Likert-type scale (1 – strongly disagree to 5 – strongly agree), and items 2 and 9 were reverse-coded before calculating an overall score. Internal consistency was acceptable pre-test (Cronbach's  $\alpha$ =0.751) but not post-test (Cronbach's  $\alpha$ =0.495), the potential reasons for which are explored in the 'Discussion' section.

# Data analysis

Quantitative analysis. Descriptive statistics were used to report rates of recruitment, retention and engagement, levels of satisfaction and perceived helpfulness of the programme, as well as pre-post scores on outcome measures. Inferential statistics were used to explore whether there were any statistically significant changes in scores on outcome measures before and after the programme. O-O plots and Shapiro-Wilk tests were first used to determine normality. Where assumptions of normality were met, a parametric paired sample t-test was used to compare pre-post scores, and where assumptions of normality were not met, the non-parametric Wilcoxon signed-rank test was used instead. Based on the recommendations for exploratory studies (Armstrong, 2014), no Bonferroni adjustments were made and p values of < 0.05 were considered statistically significant.

Outcome measures were fully completed by all participants in the pre-programme survey, but post-data was missing for one case on the DCS. This case was excluded from the analysis of pre-post changes on the DCS, but included in the analysis of other outcome measures.

Qualitative analysis. Qualitative data in the survey was analysed using qualitative content analysis (OCA), following the eight main steps outlined by Schreier (2012): (1) deciding on the research question, (2) selecting the material, (3) building a coding frame, (4) dividing the material into units of coding, (5) trying out the coding frame, (6) evaluating and modifying the coding frame, (7) main analysis and (8) interpreting and presenting findings. The first author created main categories deductively based on the survey questions (see Supplementary File 2) and then developed sub-categories inductively from the data. To ensure reliability, a second researcher independently coded all the data. Intercoder agreement on codes ranged from 78% to 100%, with only 2 out of 28 coding categories achieving less than 80% agreement (as number of codes were small, a single discrepancy resulted in a large percentage difference). All discrepancies were discussed and consensus was reached on the final coding frame and frequencies.

Following separate analysis, quantitative and qualitative data were integrated narratively in the 'Results' section, organised around the research question they answered (feasibility, acceptability or potential impact).



Figure 2. Participant flowchart.

### Community involvement

The research topic was informed by autistic people's priorities, preferences and experiences as identified from previous literature (Han et al., 2022), consultation with the community (Han et al., 2023) and direct input from two autistic team members. Our autistic team members were involved in developing and delivering the intervention, determining inclusion criteria and recruiting participants, as well as choosing and adapting outcome measures. However, they did not have direct access to survey responses, given that their dual role as facilitators may have prevented participants from critiquing the programme openly. Nonetheless, they had the opportunity to review findings after analysis and share their reflections, which are incorporated in the 'Discussion' section.

# Results

# Feasibility of the intervention

Recruitment and retention. Of 32 participants who completed the pre-programme survey and were enrolled into the programme, 19 (59%) completed the post-programme survey (Figure 2). Five participants withdrew due to lack of time and another two withdrew as they found the programme too difficult to complete on their own.<sup>4</sup> Six participants were lost to follow-up for unknown reasons (they had little/no visible engagement since the start of the programme and ultimately became uncontactable).

Participants' characteristics are presented in Table 2, with percentages calculated relative to the number of participants in each column (i.e. percentage of completers or non-completers, respectively). At first glance, these percentages may suggest that rates of completion were higher among participants who identified as male, were more highly educated and were employed. However, the small raw numbers make it difficult to draw any conclusions about differences in attrition based on visual inspection of the data and also preclude statistical comparison of the characteristics of completers versus non-completers.

*Engagement.* Reported and observed engagement with the programme was high among those who were not lost to attrition, especially with the self-help guide (Table 3). Among the minority who reported in the post-programme survey that they used but did not complete the guide, reasons included needing more time, focusing on sections of the guide that were most relevant or important to them, and the emotional burden of some of the content and exercises. These participants expressed an intention to either continue working on the guide or return to it at a more appropriate time. Reasons for not engaging with the facilitator or forum included not having the time to do so, not needing help or support, and social anxiety.

# Acceptability of the intervention

Satisfaction and perceived helpfulness. Quantitative data indicated that most participants who completed the postprogramme survey were overall satisfied with the format, content and duration of the programme (Table 4). All participants found the guide helpful, and those who used facilitator support also found it helpful. Perceived helpfulness of the forum varied, as half of those who used it found it helpful while the other half did not.

Qualitative data gave further insight into participants' experiences, as presented next.

Motivations and expectations. The most common motivation for taking part in the programme was to learn more about disclosure and 'make better informed disclosure decisions', whether in general or in relation to a specific situation: 'I was looking to disclose my autism to a new line manager and a new colleague but wanted help to decide when to disclose and how much exactly to disclose' (P21). Some participants were seeking support following a recent or late diagnosis: 'I'm fairly recently diagnosed and have not disclosed my diagnosis to many people yet' (P13). A few participants wanted to help other autistic people:

# Table 2. Participants' characteristics.

	Completers (n = 19, %)	Non-completers (n = 13, %)
Age		
25–34 years	2 (10.53%)	2 (15.38%)
35-44 years	6 (31.58%)	4 (30.77%)
45–54 years	5 (26.32%)	6 (46.15%)
55–64 years	4 (21.05%)	0 (0.00%)
65 + years	2 (10.53%)	(7.69%)
Gender identity	(	
Male	7 (36.84%)	(7.69%)
Female	12 (63.16%)	(84.62%)
Non-binary	0 (0.00%)	1 (7.69%)
Ethnicity		
White-British	13 (68.42%)	(84.62%)
Other White background	5 (26.32%)	1 (7.69%)
Middle Eastern	1 (5.26%)	0 (0.00%)
Mixed White and Asian	0 (0.00%)	1 (7.69%)
Education		
Secondary education	0 (0.00%)	(7.69%)
College/further education	1 (5.26%)	2 (15.38%)
University education	18 (94,74%)	10 (76.92%)
Employment status		
Full-time employment	10 (52.63%)	4 (30.77%)
Part-time/temporary/self-employment	7 (36.84%)	5 (38.46%)
Unemployed	1 (5.26%)	2 (15.38%)
Student	0 (0.00%)	1 (15.38%)
Retired	1 (5.26%)	1 (7.69%)
Autism diagnostic status	(0.20,0)	
Formally diagnosed	16 (84.21%)	(84.62%)
<  year	4 (21.05%)	1 (7.69%)
I-3 years	5 (26.32%)	3 (23.08%)
4–6 years	3 (15.79%)	5 (38.46%)
7–9 years	2 (10.53%)	1 (7.69%)
10 years or longer	2 (10.53%)	1 (7.69%)
Self-identified	3 (15.79%)	2 (15.38%)
<   year	1 (5.26%)	0 (0.00%)
I-3 years	1 (5.26%)	1 (7.69%)
4–6 years	0 (0.00%)	0 (0.00%)
7–9 years	0 (0.00%)	0 (0.00%)
10 years or longer	1 (5.26%)	1 (7.69%)
Openness about autistic identity/diagnosis		
Not open	0 (0.00%)	(7.69%)
Selectively open	15 (78.95%)	8 (61.54%)
Open	2 (10.53%)	3 (23.08%)
Verv open	2 (10.53%)	1 (7.69%)
Other diagnoses/conditions <sup>a</sup>	2 (10.00,0)	(())
Physical disability	2 (10.53%)	4 (30,77%)
l earning disability	4 (21.05%)	5 (38 46%)
Intellectual disability	0 (0.00%)	(7.69%)
Affective condition (e.g. depression and anxiety)	12 (63.16%)	8 (61.54%)
Behavioural condition (e.g. ADHD. Tourette syndrome)	3 (15.79%)	2 (15.38%)
Mental health condition (e.g. bipolar disorder, schizophrenia)	2 (10.53%)	3 (23.08%)

ADHD: attention-deficit hyperactivity disorder.

<sup>a</sup>The total percentages exceed 100% as some participants had multiple co-occurring diagnoses. Participants' level of openness about their co-occurring diagnoses varied, with a few noting that it was easier to talk openly about visible than non-visible conditions.

Self-reported engagement	Did not use	Used	Completed
Self-help guide	0 (0.00%)	19 (100%)	14 (73.68%)
Facilitator support	7 (36.84%)	12 (63.16%)	N/A
Online forum	5 (26.32%)	14 (73.68%)	N/A
Observed engagement	Never	Once	More than once
Emailed facilitator	4 (21.05%)	(57.89%)	4 (21.05%)
Posted on forum	5 (26.32%)	7 (36.84%)	7 (36.84%)

Table 3. Self-reported and observed engagement.

N/A: not applicable.

Table 4. Levels of satisfaction and perceived helpfulness.

Level of satisfaction	Very satisfied	Satisfied	Neither satisfied nor dissatisfied	Dissatisfied	Very dissatisfied
Overall	8 (42.11%)	9 (47.37%)	2 (10.53%)	0 (0.00%)	0 (0.00%)
Format	9 (47.37%)	8 (42.11%)	I (5.26%)	I (5.26%)	0 (0.00%)
Content	9 (47.37%)	8 (42.11%)	2 (10.53%)	0 (0.00%)	0 (0.00%)
Duration	4 (21.05%)	12 (63.16%)	I (5.26%)	2 (10.53%)	0 (0.00%)
Level of helpfulness	Very helpful	Quite heslpful	Not very helpful	Not helpful at all	N/A – did not use
Self-help guide	(57.89%)	8 (42.11%)	0 (0.00%)	0 (0.00%)	0 (0.00%)
Facilitator support	5 (26.32%)	8 (42.11%)	0 (0.00%)	0 (0.00%)	6 (31.58%)
Online forum	3 (15.79%)	5 (26.32%)	6 (31.58%)	2 (10.53%)	3 (15.79%)

N/A: not applicable.

'helping and advising others about disclosure and why it is important although [it's] the individual's decision' (P12). Many participants reported that their expectations of the programme were met: 'My hopes and aims for the programme were fully met. It has given me the tools – and the understanding – to make much better sharing decisions' (P24). Some felt that there were aspects of the programme that met their expectations and aspects that did not, as explored next.

Most liked or helpful aspects. Participants mentioned the clear structure as their favourite aspect of the programme: 'I like the way it takes me through a process of understanding - it gives me a structure in which to think it all through' (P19). They liked that the self-help guide had explanations and examples of different disclosure strategies and scenarios that were detailed and easy to understand: 'I especially liked the way that the elements of disclosure were unpacked, together with the risks and opportunities of disclosing' (P20). This explanatory and contextual information was balanced with practical exercises: 'Useful worksheets to think through in clear stages the different options, and workshopping/trying them out' (P09). Participants also appreciated that autistic peer support was 'available but optional', both via email and the forum. Even those who did not actively engage with their

facilitators or other participants acknowledged that it was 'good to hear from others' and comforting to know they could reach out if they needed help: 'I only engaged minimally, but I think it is very important to know there is an autistic person on the other end available for support' (P17). Finally, a few participants liked that the programme was self-guided, 'so there was time to think without pressure or expectations' (P07).

Least liked or challenging aspects. Many participants chose participating in the forum as their least favourite component of the programme. Some explained that they were not ready to participate because it felt 'too much like group disclosure', they were still learning about the topic or they did not have the capacity to handle negative emotions being shared on the platform: 'Became a dumping ground for negative emotions, which takes a lot of energy to engage with, and isn't a stage I'm at or need to be involved in in my own processing about disclosure' (P09). Several participants thought that the guide was too 'wordy' and found it challenging to complete the programme within the recommended time frame and/or stay motivated with selfstudy: 'structuring time to do a course like this alone can be really hard – procrastination and avoidance certainly came to play for me, especially when the topic of disclosure is such a personal and distressing thing' (P02). Finally,

a few cited personal struggles that they attributed to their own background and experiences, rather than to the programme itself: 'Not a criticism of the programme, but I am still struggling with developing a personal narrative (more to do with the complexities of my history and circumstances)' (P29).

Suggestions for improvement. A common recommendation was improving the way information was presented in the self-help guide, in particular having more interactive elements: 'I think it would be better presented as a web app, it would feel more interactive thus improve the user's experience. It could incorporate video, and give you an option for downloading your responses' (P07). Several participants also suggested conducting the programme over a longer time scale with 'more monitoring of progress' and initiation from facilitators rather than putting the onus on participants to initiate discussion: '[I] feel the peer support group could have been facilitated a bit more to guide or encourage participation' (P23). Other participants proposed additional content, such as examples of what to say in response to potential negative disclosure reactions, so that they would feel more equipped to handle stigma. A summary of the qualitative coding frame and frequencies is presented in Table 5.

Table !	5.	Qualitative	coding	frame	and	frequencies.

Impact on disclosure. Overall mean scores on the Decisional Conflict Scale (DCS) significantly decreased following the programme, as did scores on all subscales (Table 6). These findings suggest that participants felt (1) more informed, (2) clearer about personal values regarding benefits and risks, (3) more supported in decisionmaking, (4) more certain about their decision and (5) more effective at decision-making by the end of the programme. Mean scores on the adapted Disclosure-Related Distress Scale (DRDS) also increased significantly, reflecting lower levels of disclosure-related distress. Score differences on individual DRDS items showed that participants felt more comfortable disclosing to (1) a family member or friend, (2) an employer or educator and (3) a public service provider after the programme, although only changes on the first two options reached statistical significance.

Qualitative data gave additional insight on participants' perceptions of how the programme changed their feelings about disclosure. A sense of empowerment was apparent, including feeling 'more informed', 'more in control' and 'more confident' about disclosure: 'the powerful thought

Categories	Sub-categories (n=19, %)					
Motivations and	Before the programme	Help oneself with disclosure ( $n = 14, 73.68\%$ )				
expectations		Support for recent/late diagnosis (n=7, 36.84%)				
		Help others $(n=3, 15.79\%)$				
	After the programme	Expectations fully met $(n = 8, 42.11\%)$				
		Expectations partially met ( $n = 5, 26.32\%$ )				
Most liked or helpful	Availability of peer support	( <i>n</i> = 15, 78.95%)				
aspects	Clear structure ( $n = 11, 57.8$	9%)				
	Explanations and examples (	(n = 11, 57.89%)				
	Practical exercises ( $n=9, 47$	Practical exercises $(n=9, 47.37\%)$				
	Self-guided format $(n = 4, 21.05\%)$					
Least liked or challenging	The forum ( <i>n</i> = 9, 47.37%)					
aspects	Personal struggles (n=6, 31.58%)					
	Working independently $(n = 5, 26.32\%)$					
	Time pressure $(n=4, 21.05\%)$					
	Lengthy text $(n = 3, 15.79\%)$					
Suggestions for	More facilitation $(n=8, 42.11\%)$					
improvement	More interactive presentation ( $n = 5, 26.32\%$ )					
	Longer time scale (n=3, 15.79%)					
	Additional content (n = 3, 15.79%)					
Impact of the programme	Impact on disclosure	Clarify thought process $(n=9, 47.37\%)$				
		Sense of empowerment ( $n = 4, 21.05\%$ )				
		Nuanced approach (n=5, 26.32%)				
		Reaffirmed previous thoughts ( $n = 2$ , 10.53%)				
		Tool/framework to go back to $(n=7, 36.84\%)$				
		Apply learning to other 'hidden' identities ( $n = 7, 36.84\%$ )				
	Impact on stigma	Knowledge and techniques to counter stigma ( $n = 7, 36.84\%$ )				
	-	No change/still need more confidence ( $n = 5, 26.32\%$ )				
	Understanding how stigma affects self ( $n=2$ , 10					

Table 6. Pre-post change in scores on outcome measures.

Measure		Pre-intervention means (SD)	Post-intervention means (SD)	Pre-post change (Student's t/Wilcoxon W statistic, p value, effect size)
DCS	Overall	54.08 (23.92)	25.09 (15.65)	t(17) = 5.16, p < 0.001, d = 1.22
	Subscale I	57.41 (28.71)	14.81 (15.00)	W=120.00, <i>p</i> < 0.001, <i>r</i> =1.00
	Subscale 2	53.70 (29.04)	19.44 (15.12)	t(17) = 6.46, p < 0.001, d = 1.52
	Subscale 3	48.15 (22.24)	29.17 (16.73)	t(17) = 2.89, p = 0.010, d = 0.68
	Subscale 4	64.81 (27.35)	38.89 (26.51)	t(17) = 3.35, p = 0.004, d = 0.79
	Subscale 5	48.26 (25.44)	23.61 (18.26)	W=135.00, p=0.006, r=0.76
DRDS	Overall	3.16 (1.68)	4.14 (1.40)	t(18) = -2.79, p = 0.012, d = -0.64
	ltem l	3.26 (2.00)	4.11 (1.88)	W=10.50, p=0.025, r=-0.73
	ltem 2	2.68 (1.63)	4.00 (1.76)	t(18) = -3.25, $p = 0.004$ , $d = -0.75$
	Item 3	3.53 (1.95)	4.32 (1.80)	t(18) = -1.56, p = 0.135, d = -0.36
SSS	Overall	0.29 (1.41)	-0.39 (1.63)	t(18) = 2.02, p = 0.058, d = 0.46
	Subscale I	3.33 (0.89)	3.07 (1.08)	t(18) = 1.22, p = 0.237, d = 0.28
	Subscale 2	3.04 (0.84)	3.46 (0.98)	t(18) = -1.89, p = 0.075, d = -0.43
ISMI-10	Overall	2.41 (0.59)	2.27 (0.41)	t(18) = 1.52, p = 0.146, d = 0.35

DCS: Decisional Conflict Scale; DRDS: Disclosure-Related Distress Scale; SSS: Stigma Stress Scale; ISMI: Internalised Stigma of Mental Illness.

for me was to stop and think about my objective in telling someone. I think I felt that some people almost had a right to know ... But I now realise the choice is mine' (P07). Participants learned to take a more nuanced approach to disclosure: 'I [no] longer view it as binary; I can tailor disclosure according to needs and situations' (P20). Some participants felt that the programme reaffirmed previous thoughts about disclosing and had applications to their other concealable identities:

it confirmed that I should be open about my hidden disabilities (tinnitus and partial hearing loss) ... others cannot help me if they aren't aware of my difficulties. Disclosure has already resulted in my [employers] agreeing to certain adjustments at work. (P21)

Other participants spoke of how the programme helped to clarify their reasoning, even if they had not reached a final decision: 'Even though I haven't come to conclusions yet, I am much clearer in my mind about what to consider, which is actually a big relief' (P19). Participants felt that they had gained 'a tool to go back to' beyond the duration of the programme: 'It has given me a valuable framework upon which I can make future disclosure decisions' (P24).

*Impact on stigma.* There was a statistically non-significant reduction in overall mean scores on the adapted SSS, with subscale scores showing that participants perceived less harm caused by stigma and more resources to cope with stigma after the programme. There was also a non-significant reduction in overall mean scores on the ISMI-10, as shown in Table 6.

Qualitative data illuminated participants' perceptions of the impact of the programme on their approach to stigma. Some participants stated that the programme gave them knowledge and techniques to counter stigma: 'framing things more positively ... gauging people's understanding and working from there' (P29). Others felt that the programme helped them to understand how stigma affects them at an individual level: 'It's made me realise I still have some accepting to do of my diagnosis and that often the stigma that comes from other people feeds into my own anxieties about myself' (P07). A few expressed that they still needed to develop more confidence to challenge stigma: 'I would like to get to a stage where I can be an advocate and directly address stigma but need to build more confidence first' (P23).

# Discussion

We sought to assess the feasibility, acceptability and potential impact of a new disclosure decision-making programme for autistic adults using a single-arm pre-post design. Overall, recruitment of participants was successful, and we exceeded our target sample size. The programme garnered the interest of many autistic adults who were seeking support with disclosure, showing that the programme addressed a need in the community. However, attrition was high (41%), which could be attributed to a few factors. First, based on the reasons provided for withdrawing and feedback from those who completed the programme, some participants found it challenging to work through the programme independently and within the recommended time frame. Similar challenges were observed in a self-guided version of HOP for mental health professionals, where time constraints and lack of active peer support likely contributed to high dropout rates (31%) (Scior et al., 2021). Indeed, while online self-help interventions generally reach a wider audience than in-person interventions, they tend to have poorer retention as

participants are less closely supervised (Eysenbach, 2005; Geraghty et al., 2010). A self-guided format may increase the accessibility of stigma-related interventions by offering more privacy (Mills et al., 2020; Thomas et al., 2015), but our study shows that this approach may be challenging for participants who may need more practical or emotional support to engage.

Another possible factor contributing to the low retention rate could be the broad inclusion criteria we implemented for our study. Specifically, our study included all autistic adults regardless of their level of openness about their autistic identity (see Note 1 for an explanation). In contrast, some previous HOP studies more narrowly targeted individuals who had not fully disclosed or were worried about disclosing (e.g. Mulfinger et al., 2018; Scior et al., 2021). It is noteworthy that four of our six participants who were lost to follow-up for unknown reasons had indicated that they were already very open about their autistic identity before the start of the programme. While they still expressed interest in taking part after we clarified the purpose and target group of our intervention, it is possible that they realised upon starting the programme that it was indeed not relevant or helpful to them. However, as these participants did not provide a reason for withdrawing and there were no observable differences in levels of openness between completers and non-completers at baseline, no conclusions can be drawn on whether the programme is more suitable for autistic adults who are less open about their autistic identity. The only participant who reported being 'not open' at the beginning was lost to attrition, while there were four completers who were open about their autistic identity and still found the programme helpful.

Encouragingly, among those who completed the programme, levels of engagement and satisfaction were high. Reported and observed engagement with the self-help guide were higher than with the facilitators and forum. Similarly, all participants who used the guide and facilitator support found them helpful, while perceived helpfulness of the forum was mixed. Compared to online autistic communities that are larger, ongoing and with regular members (Brownlow & O'Dell, 2006; Parsloe, 2015), the forum in this programme involved a time-limited, small group of individuals who were unsure about disclosing their autistic identity. Although we hoped that a forum would feel less intrusive and anxiety-inducing than synchronous group sessions, some participants still did not feel comfortable or ready to participate. Yet, many also said that it was reassuring to know that they could get direct support from another autistic person if they needed it, affirming the value of autistic-led peer support based on shared identity and experience (Crane et al., 2021; Shea et al., 2022). Interestingly, participants also found their learning on the programme applicable to other 'hidden' identities, including physical health and mental health conditions. This observation points to the possibility of taking a more intersectional approach to the programme in the future, although it also raises the question of whether facilitators can meaningfully support participants in considering the disclosure of other identities and experiences that they may not share (Corrigan et al., 2018).

Facilitators reflected on participants' varied feedback, from those who struggled to complete the programme due to past negative experiences to those who became more confident about disclosure as they progressed through the programme, which aligned with the diversity of the autistic population. They stressed the importance of being willing to listen to participants, take feedback on board and make adjustments as a recognition of individuality, rather than imply that it is the fault of the person. Reflecting on their own experiences, they expressed that it took time to warm up to their roles and become more confident as facilitators. They empathised with participants' struggles using the forum and agreed that future iterations could include more guidance for facilitators on when to intervene or topics to discuss on the forum so they can provide more structured support to participants. Facilitators also felt that it would have been helpful for them to work through the whole guide beforehand (although they tried out parts of it during the training session) to better familiarise themselves with the content and give participants a better gauge of the time needed to complete each section.

While the primary aim of this study was not to assess the effectiveness of the programme, our preliminary results suggest that the programme has potential to improve psychosocial outcomes for autistic adults. The programme was associated with improvements in decisional conflict and disclosure-related distress, as well as reductions in stigma-related stress and internalised stigma. Qualitative data corroborated quantitative data, as participants reported that the programme gave them a greater sense of clarity, confidence, control and choice over disclosure, as well as some knowledge and techniques to counter stigma, but they still needed to work on developing self-acceptance and confidence to deal with stigma. This is understandable considering that the programme focused more on disclosure than on stigma, and unhelpful self-beliefs can be deeply ingrained and take a long time to change. Moving forward, adding booster sessions and follow-up measures may help to enhance the effects of the intervention and give insight into its long-term impact.

Most participants found the outcome measures appropriate and easy to complete (see Supplementary File 3 for examples of comments), which was supported by little missing data (among those who were not lost to attrition). However, many also pointed out that the scales contained general and binary statements that could not fully capture the nuances of disclosure taught on the programme, echoing common problems identified by autistic people with other measures (Nicolaidis et al., 2020). For example, while the DRDS asked about disclosing to a family member or friend, participants remarked that their level of comfort would differ depending on the specific person. With the ISMI-10, participants highlighted that items on social avoidance were tricky as they may avoid social situations not because of stigma but because they do not enjoy socialising. They also commented that it was difficult to answer questions about what other people thought, and they were unsure whether some questions were asking about their own beliefs or others' beliefs. It is noteworthy that the ISMI-10 was the only measure used in this study that had reverse-scored items and was also the one with low internal consistency. High levels of internal consistency on the other measures without reverse-scored items could be due to demand characteristics and participants' tendency to respond positively. Finally, as we scored the SSS and ISMI-10 on 5-point Likert-type scales for consistency in response options (instead of their original 7-point and 4-point scales, respectively), this limits the comparability of the scores in our study with other studies. In a future trial, analysis of whether greater quantitative improvement in outcomes are more apparent among those who qualitatively report a positive impact would help to provide greater evidence for the validity of measures.

Our initial findings should be interpreted with caution given the small sample size, nature of the study design and high attrition rate. Despite the high engagement and satisfaction rates among completing participants, the programme may not have been feasible and acceptable to participants who dropped out. Unfortunately, there is limited information on the latter group of participants, apart from those who provided reasons for withdrawing. As there were two participants who withdrew from the study due to reading-related difficulties, the existing textheavy nature of the programme may limit its accessibility across the spectrum. It is also a limitation that data on motivation to enrol was only collected at post-test, as a more accurate indicator might be attained at pre-test and allow exploration of whether motivations differed between completers and non-completers. Due to the small sample size, it was not possible to statistically analyse differences in retention, engagement or satisfaction based on demographic characteristics, diagnostic status or openness about autistic identity. Furthermore, our sample was highly educated and more highly employed compared to the population-level demographics of autistic people in the United Kingdom (Office for National Statistics, 2022), which limits the generalisability of our findings. Moreover, due to the absence of a control or comparison group, it is unknown if similar changes in outcomes would have occurred naturally over time or with the support of friends, family, therapists or through online forums. It is also impossible to comment on whether the 'Who, When, How to Share' programme is better than broader support programmes for autistic people that have a segment on disclosure, such as the UCLA PEERS programme (Laugeson et al., 2011) or Exploring Being Autistic programme (Crane et al., 2021).

All in all, the current study shows that further development and evaluation of the 'Who, When, How to Share' programme is warranted. The high recruitment, engagement and satisfaction rates, favourable qualitative feedback and minimal missing data suggest that the intervention and data collection procedures were generally feasible and acceptable. The positive psychosocial outcomes also provide evidence of promise regarding the intervention's impact. At the same time, participants and facilitators provided useful suggestions that can be implemented to improve the intervention, strengthen the evaluation and increase participant retention. First, more audio and visual materials should be incorporated to motivate greater attention and make the programme more accessible to those who find it challenging to read lengthy documents. Second, while this study had a restricted time frame for data collection purposes, a more flexible timeline may remove a barrier to completion, although this would need to be accompanied by closer monitoring of participants' progress to prevent feelings of disengagement. Third, while our facilitators adopted a 'light touch' approach to moderating the forum, amplifying the role of facilitators in leading interactive exercises or providing prompt questions may promote greater engagement. A sufficiently powered randomised controlled trial would be needed to evaluate the effectiveness of this programme and examine factors that may be associated with variation in outcomes.

#### **Declaration of conflicting interests**

The author(s) declared the following potential conflicts of interest with respect to the research, authorship and/or publication of this article: The authors disclosed their involvement in both the development and evaluation of the programme reported in this article. They declare no other conflicts of interest.

# Funding

The author(s) disclosed receipt of the following financial support for the research, authorship and/or publication of this article: Funding for this study was provided by University College London (UCL).

# Ethics

Ethical approval for this study was obtained via the Department of Psychology and Human Development at IOE, UCL's Faculty of Education and Society. Informed consent for participation and publication was obtained from all study participants.

# **ORCID** iDs

Emeline Han D https://orcid.org/0000-0002-4955-7112 Kana Grace D https://orcid.org/0000-0002-6879-7779 Simone Dufresne D https://orcid.org/0000-0003-2880-7917 Laura Crane D https://orcid.org/0000-0002-4161-3490

### **Data accessibility**

Ethical approval and participant consent were not obtained for raw data sharing. Due to the confidential and sensitive nature of our data, there is no data that can be disclosed beyond that contained within the manuscript and supplemental material.

### Supplemental material

Supplemental material for this article is available online.

### Notes

- 1. As the programme focuses on disclosure decision-making, it was originally envisaged that it may be more suitable for autistic adults with a formal diagnosis and who had not fully disclosed. Indeed, our consultation survey suggested that not all autistic adults might need such a programme or find it helpful, such as those who are already confident about disclosure (Han et al., 2023). For similar reasons, some previous HOP (Honest, Open, Proud) studies have used pre-existing levels of disclosure as an inclusion criterion (e.g. Scior et al., 2021). However, our autistic team members were concerned that requiring a formal diagnosis may be seen as 'gatekeeping' and reinforcing the lack of support available for undiagnosed autistic adults. Furthermore, during the recruitment process, a number of prospective participants who were openly autistic still expressed an interest in the programme. We included these participants as long as they provided a reason for participating that was broadly in line with the goal of our programme, showing that they understood it and thought it would be beneficial to them. For example, some felt that they had been too open in the past and wanted to become more selective and strategic with disclosure.
- Between the first and second iteration, minor adjustments were made to the programme materials based on participant feedback to improve accessibility, namely increasing font size and spacing, decreasing colour contrast and enabling text-to-speech and speech-to-text functions. All other aspects remained the same.
- 3. Reliability assessment of Cronbach's alpha values followed the rules of thumb provided by George and Mallery (2003):  $\alpha > 0.9$  is excellent,  $\alpha > 0.8$  is good,  $\alpha > 0.7$  is acceptable,  $\alpha > 0.6$  is questionable,  $\alpha > 0.5$  is poor and  $\alpha < 0.5$  is unacceptable.
- 4. One participant disclosed dyslexia and the other disclosed general difficulties in reading and comprehension. After minor adjustments were made to the self-help guide based on their feedback (see Note 2 for details), both participants were offered the opportunity to try out the materials again, but neither responded.

### References

- Armstrong, R. A. (2014). When to use the Bonferroni correction. Ophthalmic and Physiological Optics, 34(5), 502–508. https://doi.org/10.1111/opo.12131
- Botha, M., & Frost, D. M. (2020). Extending the minority stress model to understand mental health problems experienced by the autistic population. *Society and Mental Health*, *10*(1), 20–34. https://doi.org/10.1177/2156869318804297

- Boyd, J. E., Otilingam, P. G., & Deforge, B. R. (2014). Brief version of the Internalized Stigma of Mental Illness (ISMI) Scale: Psychometric properties and relationship to depression, self esteem, recovery orientation, empowerment, and perceived devaluation and discrimination. *Psychiatric Rehabilitation Journal*, 37(1), 17–23. https:// doi.org/10.1037/pri0000035
- Brownlow, C., & O'Dell, L. (2006). Constructing an autistic identity: AS voices online. *Mental Retardation*, 44(5), 315– 321. https://doi.org/10.1352/0047-6765(2006)44[315:CAA IAV]2.0.CO;2
- Cage, E., & Troxell-Whitman, Z. (2019). Understanding the reasons, contexts and costs of camouflaging for autistic adults. *Journal of Autism and Developmental Disorders*, 49(5), 1899–1911. https://doi.org/10.1007/s10803-018-03878-x
- Cook, J., Hull, L., Crane, L., & Mandy, W. (2021). Camouflaging in autism: A systematic review. *Clinical Psychology Review*, 89, 102080. https://doi.org/10.1016/j.cpr.2021.102080
- Corrigan, P. W., Kosyluk, K. A., & Rüsch, N. (2013). Reducing self-stigma by coming out proud. *American Journal of Public Health*, 103(5), 794–800. https://doi.org/10.2105/ AJPH.2012.301037
- Corrigan, P. W., Nieweglowski, K., Buchholz, B. A., & Al-Khouja, M. A. (2017). Honest, Open, Proud to eliminate the stigma of mental illness. Workbook for program facilitators and participants. https://hopprogram.org/
- Corrigan, P. W., Rüsch, N., & Scior, K. (2018). Adapting disclosure programs to reduce the stigma of mental illness. *Psychiatric Services*, 69(7), 826–828. https://doi. org/10.1176/appi.ps.201700478
- Crane, L., Hearst, C., Ashworth, M., Davies, J., & Hill, E. L. (2021). Supporting newly identified or diagnosed autistic adults: An initial evaluation of an autistic-led programme. *Journal of Autism and Developmental Disorders*, 51(3), 892–905. https://doi.org/10.1007/s10803-020-04486-4
- Eysenbach, G. (2005). The law of attrition. Journal of Medical Internet Research, 7(1), Article e11. https://doi.org/10.2196/ jmir.7.1.e11
- Flower, R. L., Dickens, L. M., & Hedley, D. (2021). Barriers to employment: Raters' perceptions of male autistic and nonautistic candidates during a simulated job interview and the impact of Diagnostic disclosure. *Autism in Adulthood*, 3(4), 300–309. https://doi.org/10.1089/aut.2020.0075
- George, D., & Mallery, P. (2003). SPSS for Windows step by step: A simple guide and reference, 11.0 update (4th ed.). Allyn & Bacon.
- Geraghty, A. W. A., Wood, A. M., & Hyland, M. E. (2010). Attrition from self-directed interventions: Investigating the relationship between psychological predictors, intervention content and dropout from a body dissatisfaction intervention. *Social Science & Medicine*, 71(1), 30–37. https://doi. org/10.1016/j.socscimed.2010.03.007
- Gillespie-Lynch, K., Bisson, J. B., Saade, S., Obeid, R., Kofner, B., Harrison, A. J., Daou, N., Tricarico, N., Delos Santos, J., Pinkava, W., & Jordan, A. (2022). If you want to develop an effective autism training, ask autistic students to help you. *Autism*, 26(5), 1082–1094. https://doi. org/10.1177/13623613211041006
- Han, E., Scior, K., Avramides, K., & Crane, L. (2022). A systematic review on autistic people's experiences of stigma and

coping strategies. *Autism Research*, 15(1), 12–26. https://doi.org/10.1002/aur.2652

- Han, E., Scior, K., Heath, E., Umagami, K., & Crane, L. (2023). Development of stigma-related support for autistic adults: Insights from the autism community. *Autism*, 27(6), 1676– 1689. https://doi.org/10.1177/13623613221143590
- Huang, Y., Hwang, Y. I., Arnold, S. R., Lawson, L. P., Richdale, A. L., & Trollor, J. N. (2022). Autistic adults' experiences of diagnosis disclosure. *Journal of Autism and Developmental Disorders*, 52(12), 5301–5307. https://doi.org/10.1007/ s10803-021-05384-z
- Jones, D. R., DeBrabander, K. M., & Sasson, N. J. (2021). Effects of autism acceptance training on explicit and implicit biases toward autism. *Autism*, 25(5), 1246–1261. https://doi. org/10.1177/1362361320984896
- Julious, S. (2005). Sample size of 12 per group rule of thumb for a pilot study. *Pharmaceutical Statistics*, *4*(4), 287–291. https://doi.org/10.1002/pst.185
- Kapp, S. K., Gillespie-Lynch, K., Sherman, L. E., & Hutman, T. (2013). Deficit, difference, or both? Autism and neurodiversity. *Developmental Psychology*, 49(1), 59–71. https://doi. org/10.1037/a0028353
- Lancaster, G., Dodd, S., & Williamson, P. (2004). Design and analysis of pilot studies: Recommendations for good practice. *Journal of Evaluation in Clinical Practice*, 10(2), 307– 312. https://doi.org/10.1111/j..2002.384.doc.x
- Laugeson, E. A., Frankel, F., Gantman, A., Dillon, A. R., & Mogil, C. (2011). Evidence-based social skills training for adolescents with autism spectrum disorders: The UCLA peers program. *Journal of Autism and Developmental Disorders*, 42(6), 1025–1036. https://doi.org/10.1007/ s10803-011-1339-1
- Love, A. M. A., Edwards, C., Cai, R. Y., & Gibbs, V. (2023). Using experience sampling methodology to capture disclosure opportunities for autistic adults. *Autism in Adulthood*. Advance online publication. https://doi.org/10.1089/ aut.2022.0090
- McMahon, C. M., Henry, S., & Linthicum, M. (2021). Employability in autism spectrum disorder (ASD): Job candidate's diagnostic disclosure and ASD characteristics and employer's ASD knowledge and social desirability. *Journal* of Experimental Psychology: Applied, 27(1), 142–157. https://doi.org/10.1037/xap0000282
- Mills, H., Mulfinger, N., Raeder, S., Rüsch, N., Clements, H., & Scior, K. (2020). Self-help interventions to reduce selfstigma in people with mental health problems: A systematic literature review. *Psychiatry Research*, 284, 112702. https:// doi.org/10.1016/j.psychres.2019.112702
- Mulfinger, N., Müller, S., Böge, I., Sakar, V., Corrigan, P. W., Evans-Lacko, S., Nehf, L., Djamali, J., Samarelli, A., Kempter, M., Ruckes, C., Libal, G., Oexle, N., Noterdaeme, M., & Rüsch, N. (2018). Honest, Open, Proud for adolescents with mental illness: Pilot randomized controlled trial. *Journal of Child Psychology and Psychiatry*, 6, 684–691. https://doi.org/10.1111/jcpp.12853
- Nicolaidis, C., Raymaker, D. M., McDonald, K. E., Lund, E. M., Leotti, S., Kapp, S. K., Katz, M., Beers, L. M., Kripke,

C., Maslak, J., Hunter, M., & Zhen, K. Y. (2020). Creating accessible survey instruments for use with autistic adults and people with intellectual disability: Lessons learned and recommendations. *Autism in Adulthood*, *2*(1), 61–76. https://doi.org/10.1089/aut.2019.0074

- O'Connor, A. (1995). Validation of a decisional conflict scale. *Medical Decision Making*, 15(1), 25–30. https://doi.org/10. 1177/0272989X9501500105
- Office for National Statistics. (2022, February 10). Outcomes for disabled people in the UK: 2021. https://www.ons.gov. uk/peoplepopulationandcommunity/healthandsocialcare/ disability/articles/outcomesfordisabledpeopleintheuk/2021
- Parsloe, S. M. (2015). Discourses of disability, narratives of community: Reclaiming an autistic identity online. *Journal of Applied Communication Research*, 43(3), 336–356. https:// doi.org/10.1080/00909882.2015.1052829
- Romualdez, A. M., Heasman, B., Walker, Z., Davies, J., & Remington, A. (2021). 'People might understand me better': Diagnostic disclosure experiences of autistic individuals in the workplace. *Autism in Adulthood*, 3(2), 157–167. https://doi.org/10.1089/aut.2020.0063
- Rüsch, N., Corrigan, P. W., Wassel, A., Michaels, P., Olschewski, M., Wilkniss, S., & Batia, K. (2009). A stress-coping model of mental illness stigma: I. Predictors of cognitive stress appraisal. *Schizophrenia Research*, 110(1–3), 59–64. https://doi.org/10.1016/j.schres.2009.01.006
- Rüsch, N., & Kösters, M. (2021). Honest, Open, Proud to support disclosure decisions and to decrease stigma's impact among people with mental illness: Conceptual review and meta-analysis of program efficacy. Social Psychiatry and Psychiatric Epidemiology, 56(9), 1513–1526. https://doi. org/10.1007/s00127-021-02076-y
- Sasson, N. J., & Morrison, K. E. (2019). First impressions of adults with autism improve with diagnostic disclosure and increased autism knowledge of peers. *Autism*, 23(1), 50–59. https://doi.org/10.1177/1362361317729526
- Schreier, M. (2012). *Qualitative content analysis in practice*. Sage.
- Scior, K., Evans, J., Hildebrand, A., Mills, H., Raeder, S., Smith, V., Corrigan, P. W., & Clements, H. (2021). A guided selfhelp intervention supporting mental health professionals' decisions regarding sharing of lived experience. *Clinical Psychology Forum*, 2021(344), 65–70.
- Scior, K., Rüsch, N., White, C., & Corrigan, P. (2020). Supporting mental health disclosure decisions: The Honest, Open, Proud programme. *The British Journal of Psychiatry*, 216(5), 243–245. https://doi.org/10.1192/bjp.2019.256
- Shea, L. L., Wong, M.-Y., Song, W., Kaplan, K., Uppal, D., & Salzer, M. S. (2022). Autistic-delivered peer support: A feasibility study. *Journal of Autism and Developmental Disorders*. Advance online publication. https://doi. org/10.1007/s10803-022-05816-4
- Skivington, K., Matthews, L., Simpson, S. A., Craig, P., Baird, J., Blazeby, J. M., Boyd, K. A., Craig, N., French, D. P., McIntosh, E., Petticrew, M., Rycroft-Malone, J., White, M., & Moore, L. (2021). A new framework for developing and evaluating complex interventions: Update of Medical

Research Council guidance. *BMJ*, *374*, Article n2061. https://doi.org/10.1136/bmj.n2061

- Thomas, N., McLeod, B., Jones, N., & Abbott, J.-A. (2015). Developing internet interventions to target the individual impact of stigma in health conditions. *Internet Interventions*, 2(3), 351–358. https://doi.org/10.1016/j.invent.2015.01.003
- Thompson-Hodgetts, S., Labonte, C., Mazumder, R., & Phelan, S. (2020). Helpful or harmful? A scoping review of perceptions and outcomes of autism diagnostic disclosure to

others. Research in Autism Spectrum Disorders, 77, 101598. https://doi.org/10.1016/j.rasd.2020.101598

- Walker, N. (2021). Neuroqueer heresies: Notes on the neurodiversity paradigm, autistic empowerment, and postnormal possibilities. Autonomous Press.
- Wisconsin Initiative for Stigma Elimination & Corrigan, P. W. (2018). Up to me: Talking about our mental health? Brief Reflection Workbook: Teen. https://eliminatestigma.org/upto-me/